Author's response to reviews

Title: Fluoroquinolone Resistance During 2000-2005: An Observational Study

Authors:

Richard J Ryan (ryanrj@ucmail.uc.edu)
Christopher J Lindsell (lindsecj@ucmail.uc.edu)
Paul M Sheehan (sheehapm@uc.edu)

Version: 5 Date: 7 May 2008

Author's response to reviews:

Removed "Title" from title of manuscript
Changed authors to names and then affiliations as described in email
Removed qualifications from authors' names
Added "United States of America" after address of institution
Removed page break after acknowledgments
Removed figure title from Figure 1 and placed title after reference list before the tables.